Author's response to reviews

Title: Does the principle of minimum work apply at the carotid bifurcation: a retrospective cohort study.

Authors:

Richard J Beare (Richard.Beare@monash.edu)
Gita Das (das_gita@hotmail.com)
Mandy Ren (mren4@student.monash.edu)
Winston Chong (Winston.Chong@southernhealth.org.au)
Matthew D Sinnott (Matthew.Sinnott@csiro.au)
James E Hilton (James.Hilton@csiro.au)
Velandai Srikanth (velandai.srikanth@monash.edu)
Thanh G Phan (Thanh.Phan@monash.edu)

Version: 7 Date: 10 August 2011

Author's response to reviews:

Author initials have been added where appropriate.

Author names and formats are the same in manuscript and submissions system.

Thanh G Phan is listed as the corresponding author in the manuscript. RJ Beare will remain contact author for publication process.

All authors have seen and approved the manuscript.

references to "Supplementary" material have been changed to "additional" material.

Change tracking has been removed.

Vertical lines have been removed from tables.

Figures 2,3 and 4 have been cropped.